Author’s response to reviews

Title: Extrarenal nephroblastomatosis in children: A report of two cases

Authors:

Yi Wu (170330856@qq.com)
Xue ming Zhu (xueming_zhu@aliyun.com)
Xing dong Wang (13914073627@163.com)
Hang zhou Wang (wanghangzhoudr@126.com)
Xu Cao (cao_xu1982@163.com)
Jian Wang (wj196312@vip.163.com)

Version: 4
Date: 26 September 2014

Author’s response to reviews: see over
Dear Editor:

Thank you for your letter and for the comments concerning our manuscript entitled “Extrarenal nephroblastomatosis in children: A report of two cases” (MS: 1531012475133330). These comments are all valuable and helpful for revising and improving our paper. We have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the comments are as flowing:

1. Case Presentation
   Please include information about the patient’s ethnic background in your case presentation. This should also be included in your abstract. Please see the following links for formatting these sections.
   We have described the ethnic background of the two cases in case presentation (Line 58 and Line 84) and abstract (Line 25). Both of the patients are Han Chinese of Asian.

2. Copyediting:
   We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service.
   We had this manuscript copyedited by an English speaking professional with science background at Elixigen Corporation (Huntington Beach, California, USA).

   Thank you very much for your good comments. We tried our best to improve the manuscript and made some changes. These changes will not influence the content and framework of the paper.

   We appreciate for Editors’ warm work and hope that the correction will meet with approval.

Sincerely,

Xueming Zhu
Department of Pathology, Soochow University Affiliated Children's Hospital, Jiangsu 215003, China.
Address: 303 Jingde Road, Suzhou, Jiangsu, 215003, China
E-mail: xueming_zhu@aliyun.com
Tel: +86 0512 67786712
Fax: +86 0512 65224492